Author's response to reviews

Title: ‘Simulation-based learning in Psychiatry for undergraduates at the University of Zimbabwe medical school’

Authors:

Angharad E Piette (aepiette@doctors.org.uk)
Florence Muchirahondo (fchikwanha@gmail.com)
Walter Mangezi (wmangezi@yahoo.co.uk)
Frances Cowan (f.cowan@ucl.ac.uk)
Hugh Grant-Peterkin (hughgp@gmail.com)
Amy Iversen (amy.iversen@kcl.ac.uk)
Melanie Abas (melanie.abas@kcl.ac.uk)

Version: 2
Date: 27 January 2014

Author's response to reviews:

Dear BMC Medical Education Editorial Board,

Many Thanks for your response to our submission requesting some revisions for the manuscript 'Simulation-based learning in Psychiatry for undergraduates at the University of Zimbabwe medical school'. Please find attached our updated manuscript with the additions of 'Competing Interests' and 'Acknowledgments' sections at the end of the manuscript.

We look forward to receiving further comments on our manuscript if forwarded for peer review.

Kind Regards,

Dr Walter Mangezi